Author's response to reviews

Title: Effectiveness of two interventions based on improving patient-practitioner communication on diabetes self-management in patients with low educational level. Study protocol of a clustered randomized trial in Primary Care.

Authors:

Ignacio Ricci-Cabello (ignacio.riccicabello@phc.ox.ac.uk)
Antonio Olyr de Labry-Lima (antonio.olry.easp@juntadeandalucia.es)
Julia Bolívar-Muñoz (julia.bolivar.easp@juntadeandalucia.es)
Guadalupe Pastor-Moreno (gpastormoreno@yahoo.es)
Clara Bermudez-Tamayo (clara.bermudez.easp@juntadeandalucia.es)
Isabel Ruiz-Pérez (isabel.ruiz.easp@juntadeandalucia.es)
Fernán Quesada-Jiménez (quesadaj@meditex.es)
Enrique Moratalla-López (enmolo1976@gmail.com)
Susana Domínguez-Martín (susandominguezm@gmail.com)
Ana M De los Rios-Álvarez (anamdelosrios@yahoo.es)
Pilar Cruz-Vela (picruvel@hotmail.es)
Miguel A Padros-Quel (mapk3@hotmail.com)
José A López- De Hierro (jandreshierro@gmail.com)

Version: 2 Date: 8 March 2013

Author's response to reviews: see over
Response to comments of the Reviewers and/or Editor

Manuscript ID: MS: 6819805989266434

Title: Effectiveness of two interventions based on improving patient-practitioner communication on diabetes self-management in patients with low educational level. Study protocol of a clustered randomized trial in Primary Care.

Dear Armee Valencia,

Thank you for the opportunity to revise our above-referenced paper. After reviewing in detail the original manuscript, we have incorporated the necessary changes. The following letter gives our responses to the comments as requested in your letter of 5-Mar-2013.

Requesting TRN. We notice that you are reporting a clinical trial but have not cited a trial registration number. This must be obtained before we can begin peer review of your manuscript.

We have included the registration number of the trial on page 3, after the Abstract and the application on line.

Requesting name of ethics committee: Please update your ethics statement to include the name of the ethics committee that approved your study.

This study was approved by the Research Committee of Escuela Andaluza de Salud Pública. We have included the name of the ethics committee on page 7 of the manuscript.

Yours sincerely,

Isabel Ruiz-Perez,
Escuela Andaluza de Salud Pública
Campus Universitario de Cartuja, Cuesta del Observatorio, 4. Apartado de Correos 2070, E-18080 Granada, Spain.
Telephone number +34-670944904, fax +34-958-027503.
E-mail: isabel.ruiz.easp@juntadeandalucia.es